2006-02-24 Integration Scheduling Meeting

Friday, February 24, 2006
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Notes from Integration:
©  Third component failure in three month
© Moved beam off time to Saturday/Sunday midnight.
© Some machine studies and documentation.
Machine Summaries:
© Linac
=  Summary:
o Running well
o Thursday, readback with quad power supply
o Doing some nothing studies.
= Requests:
[m}
© Booster
=  Summary:
o Running Well
= Requests:
[m}
© Main Injector
=  Summary:
o Lambertson scans done
o Documentation studies done.
o BPMs timing in using boomerang mode. In progress
o Correctors off at 120 GeV orbit needed.

= Requests:
o 11 batch study to NuMI and Pbar.
° Pbar
=  Summary:
o Documentation studies.
o 70% done, will finish this afternoon.
= Requests:
[m}
©  Tevatron
=  Summary:
o Store 4666 had success getting better Pbar lifetime
o Quench at F4.
o Failed 200 module on abort, sloppy abort.
o Not sure why so bad a quench at F4. Have aborted cleanly
there in the past.
[m}
= Requests:
[m}
© Recycler
=  Summary:
o Store 4666 - shot from 260s.
Extracted very dense beam.
leV-sec /per bunch with transverse at 4pi.
Studies afterwards. Documenting upper and lower tune points.
e- cool
Boomerang mode - help MI commission BPM system
Will cool the remaining pbars
Keep remaining pbars through the night
o Proton studies (low intensity) on Saturday.
= Requests:
[m}

O0o0o0oooaQ

o SY120
=  Summary:
o Mcenter - taking at different energy
o Mtest- three groups
o Should get everything done by end of the run since have 1 spill
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O

per 60 sec.
= Requests:
[m}
MiniBooNE

=  Summary:
o Special run with horn at various current.s
= Requests:
o Another special run with horn at different currents.
NuMI
=  Summary:
o Horn ground current investigation - EE experts re-routed ground
loops, etc....
o Update the grounding during the shutdown.
= Requests:
[m}
[m}
CDF
=  Summary:
4666 - useful store.
179e30 initial CDF luminosity
Runiib trigger
Supervised access today
Unstacking steal from around the quads.
o Should be ready for survey on Monday of beamlines.
= Requests:

O0Oooao

O
DO
=  Summary:
o 4666 -
o 139 initial luminosity - smaller than hoped for to test trigger

table.

o Survey of detector

o In supervised accesss

Long Shutdown - Bob Mau
=  Summary:

o No access unless pre-approved before 0801 Monday, excepet
for CDF and DO

o Tentative Schedule

2/20/05 MONDAY TUESDAY WEDNESDAY THURSDAY FRIDAY SATURDAY Sunday
2/20/06 11 2/22. 23 124 2115 2126 .
Owls
Draft
to
0800 Stacking and
store
DAYS 0800 0800 Booster will
[Eooster beam e dialed down to 1
0300 ower will be turn for numi and
to dialed down from 1 turn for
1600 500 watts beam mini- boone,
ower to 200 watts
Stacking and
store
EVES 2000 ne stacking
jafter this time
1600
to Stacking and
400 |7 store
(2200 ne new shet
lfter this time

Schedule can be found at hitp fwww-bd fral gow/ operations/schedules html

Pasted from <http://www-bdnew.fnal.gov/operations/schedule/February 4 2006.gif>
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pdated MONDAY TUESDAY WEDNESDAY THURSDAY FRIDAY SATURDAY Sunday
/2105 2/27/2006 212806 3/106 3/2/06 310306 3104106 30506
Owls 001 Beam off in
0000 All machines 0600 bossterRad D
to except for TEV  urvey
0800 and Lina¢ [Eooster door cores
IChanged out
Beam off in Tev at
600
DAYS  [NTF-PT INTE-FT (Start of Linag 7 da
0800-ngon Fhar radl0800 booster ready shutdoven 0700-1700
0800 survey. [For supervised 0815-1130 1i62  Feeder 42 CUB | ~rg
to laccess 0815-1130 Mi 10 0B15-1400 1 52 i 65 [Power outage INfechamical
1600 0800 rad survey Foumi rad survey (8 MI 20 power, (8 Mi 60 power (also no power in - <
V-2 line 1000 Morning, IMil2 outages 18 hrs. 3 roor) [Maintance
supervised access [Back up gencratoy
0815-1130 Mi 30, [1300-1600 BGSO thnd 15tonaic  [[eat still
[0 jobs requiring  [Wi31 [Pwr outages lconditienar onto [ .
safety coverage rovide some )1onc_led
allowed I Mini boene rad cooling for == [T'0 Wilson
i rad survey lsurvey. [Tex road shutdown [gallery computer [yal]
500-1600 1ey rad  |And door core [Ar AZa2, A1, F4, 00T Computers Bgalte
sx""‘?y’dm”m change [F3 1building ata ack 1800 till puters off
0815113004 40 ime( may have to 0800 Sunday!  |a0n_no00
[drive long way
CwWer outages
laround
foon Ebee [Heat still provide
supervised access e Tritenn hall
EVES 1600 Tew Ready for|1700 Mi ready for ICUB turned back
Supervised access  |eupervised acce ss loss at 2000
1600 ossible dlrs of fire alarm
to orns blasting in
2400 i & Bgrev line
funnel
Schedule can be found at hitp ol Fnal Bt

Pasted from <http://www-bdnew.fnal.gov/operations/schedule/February 5 2006.gif>

[m}
= Requests:
[m}
© Mechanical
=  Summary:
[m}
= Requests:
[m}
o Other
=  Summary:
[m}
= Requests:
[m}
The Plan
o  Summary:

= 38 hours of low intensity operations.

© Requests:

Integration Meetings Page 4




